



Article Evaluation of Social Responsibility of Major Municipal Road Infrastructure—Case Study of Zhengzhou 107 Auxiliary Road Project

Delei Yang¹, Jiawen Li¹, Jiudong Peng², Jun Zhu^{3,*} and Lan Luo⁴

- ¹ School of Construction Management and Real Estate, Henan University of Economics and Law, Zhengzhou 450016, China; deleiyang_tju@126.com (D.Y.); jiawen_li@163.com (J.L.)
- ² Henan Urban Planning Institute & Corporation, Zhengzhou 450053, China; ghydqyfy@163.com
- ³ Department of Construction Management and Real Estate, Tongji University, Shanghai 200092, China
- ⁴ School of Engineering and Construction, Nanchang University, Nanchang 330031, China; mengling2391@163.com
- * Correspondence: johnsonzhu@tongji.edu.cn

Abstract: Social responsibility plays an important role in the sustainable development of major municipal road infrastructure. In this study, a major municipal road infrastructure social responsibility (MMRISR) evaluation indicator system is developed for the comprehensive evaluation of social responsibility. Questionnaires and expert interviews were used to screen the initial indicators of the proposed system. Then, 24 indicators were selected from four dimensions to establish an MMRISR evaluation indicator system. The fuzzy analytic hierarchy process was employed to calculate the weights of each indicator. Finally, the Zhengzhou 107 Auxiliary Road Project was adopted as a case study to test the reliability of the proposed evaluation system. The contribution of this study lies in the provision of a novel indicator system for the social responsibility evaluation of major municipal road infrastructures, thus improving the science of project establishment and decision-making. The proposed social responsibility system can provide an efficient decision-making tool for social responsibility governance, fundamentally promoting the sustainable development of major municipal road infrastructures and the achievement of certain sustainable development goals.

Keywords: major municipal road infrastructure; megaproject social responsibility; indicator system; fuzzy analytic hierarchy process; sustainable development

1. Introduction

The recent emergence of megaprojects has become a global phenomenon. Megaprojects, such as high-speed railroads, large expressways networks, natural gas pipeline projects, large-span bridges, and large hydropower projects, are major infrastructure projects (MIPs) [1]. MIPs considerably affect national and regional economies, public goods, the environment, politics, and many other aspects of the project lifecycle [2–6]. The continuous progress and expansion of MIPs increasingly requires a consideration of various economic, environmental, and social issues [7–9]. Given the attribute of megaprojects being public goods, their externalities should consider and fulfill social responsibilities [10,11]. Social responsibility plays a key role in improving MIP sustainability and enhancing project performance [12,13]. Megaproject social responsibility (MSR) refers to "the policies and practices of the stakeholders through the whole project lifecycle that reflect responsibilities for the well-being of the wider society" [14]. MSR has become a key factor affecting the sustainable development of MIPs [15]. As a key element of sustainable development, MSR is gradually becoming a new research hotspot in the field of engineering management [1,14]. MSR has also attracted attention from both academics and practitioners [16–18].

Major municipal road infrastructure (MMRI) is defined as the contributive component in MIPs and public service systems [14]. MMRI plays a crucial role in the development of



Citation: Yang, D.; Li, J.; Peng, J.; Zhu, J.; Luo, L. Evaluation of Social Responsibility of Major Municipal Road Infrastructure—Case Study of Zhengzhou 107 Auxiliary Road Project. *Buildings* **2022**, *12*, 369. https://doi.org/10.3390/ buildings12030369

Academic Editors: Tao Wang, Jian Zuo, Hanliang Fu and Zezhou Wu

Received: 16 February 2022 Accepted: 14 March 2022 Published: 17 March 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). urbanization and the economy at the present stage [19–21]. In general, MIPs are characterized by basic, nontradable, indivisible, and quasipublic goods, with huge investment and construction scale, wide coverage, complex stakeholders, high dynamic sensitivity, and extensive influence [1,14,22]. In addition to the above characteristics, the most significant feature of MMRI is that it is a "public good" road, which is provided to the public free of charge after project completion. The government and municipal departments are responsible for the maintenance and management of MMRIs. In addition, MMRI directly serves the inner city. Typically located in dense areas that surround residents, MMRI is closely related to the people's daily lives and travels [23,24]. However, numerous problems, including urban dust pollution, noise pollution, and soil and water damage, can easily arise in the presence of MMRIs.

The fulfillment of social responsibility can improve the performance of MMRIs in terms of their environmental, social, economic, and legal aspects [2]. MSR is an increasingly important area in MIPs' sustainable development [7,12,13,25]. Recent developments in the field of sustainability have led to the popularization of major municipal road infrastructure social responsibility (MMRISR), an aspect that is extremely difficult to ignore [14]. The lack of MMRISR negatively impacts a range of other factors, including social development, economic construction, livelihood improvement, and the achievement of sustainable development goals [2,26,27]. For example, the Karakum Canal and the Three Gorges Dam have both caused catastrophic ecosystem degradation. The Three Gorges Dam, the world's largest hydroelectric project and a symbol of China's development of MIPs, has also been fiercely criticized for the threats it poses to the environment, animal species, and migrants [16,28–30]. Therefore, there is an urgent need for MMRISR indicators to improve the fulfillment of MSR. First, existing social responsibility indicator systems usually target corporate social responsibility (CSR) [31], and they cannot fully meet the requirements of MSR [32]. Second, other stakeholders are often not considered and are excluded, except for construction parties. Third, the existing indicators mainly focus on megaprojects [2]. These indicators cannot accurately reflect the specific performance of MSR in a single type of megaprojects. Fourth, existing indicators focus on macrolevel outcomes [25], making it difficult to translate strategic goals into practical MSR actions.

Motivated by the aforementioned discussion, this study proposes an indicator system for evaluating MMRISR. The proposed system includes four dimensions, namely, the political; economic and quality; legal; and environmental and ethical dimensions, with a total of 24 secondary indicators. The fuzzy analytic hierarchy process (FAHP), a quantitative method for dealing with uncertainties and fuzziness in complex problems [23], is employed in this study to build the evaluation indicator system. MMRISR evaluation is an ambiguous and multidimensional complex problem [2,32], and the FAHP approach can be fully applied to this topic. The Zhengzhou 107 Auxiliary Road Project is analyzed as a case study to test the feasibility and practicability of the evaluation indicator system.

The indicator system is designed to respond to the question: "How do we evaluate MMRISR?" This study, which encompasses a holistic view of research, aims to establish a comprehensive scientific evaluation indicator system and provide a decision-making tool for socially responsible governance [33]. This study can fill a gap in MMRISR research, hoping to significantly promote project performance while ensuring the sustainable development of MIPs. An evaluation indicator system of MMRISR based on the FAHP approach is established, thus providing theoretical support for the sustainability evaluation of megaprojects and contributing to the achievement of sustainable development goals.

The rest of this paper is structured as follows. Section 2 describes the research methodology for establishing the MMRISR indicator system. Section 3 presents the results and analysis. Section 4 discusses the indicator system and weighting results. Section 5 summarizes the conclusions and presents the limitations of the study and the potential for future research.

2. Literature Review

Sustainability is an extremely important topic in the field of social responsibility from the viewpoint of infrastructure management [13,14,34–36]. Social responsibility in the implementation of MIPs has been investigated from a variety of perspectives. Given the advances in MSR research, there is an urgent need for methods and techniques that enable sustainability assessment and decision-making on important issues of social responsibility. Current research on social responsibility evaluation methods is not suitable for MMRISR due to the different perspectives on social responsibility and the different types of megaprojects. A comprehensive and scientific system of indicators for a single type of megaprojects is urgently needed.

First, indicators originating from the corporate perspective point only to CSR and focus only on aspects of CSR [32,37]. Many scholars, experts, and organizations have re-searched and developed various types of indicator systems, ranging from ethics and morality, to bringing charity to safety issues [38,39]. At the same time, international standards organizations have established international management standards targeting CSR with the aim of improving the motivation of companies to fulfill CSR in order to achieve better social performance, such as ISO 14001, ISO 26000, and SA 8000.

Second, the existing social responsibility evaluation indicator system is only applicable to construction companies [32,40,41]. In this indicator system, most of the stakeholders of megaprojects, such as designers, the public, and decision makers, are excluded and not considered.

Third, from the perspective of project types, these indicators only focus on the overall aspect of megaprojects. There is noticeably poor coverage of the single type of megaprojects. The MSR evaluation conclusions derived from these indicators alone do not accurately reflect the performance status and issues of MSR for a single megaproject type throughout the project life cycle. Since megaprojects involve a variety of fields, different types of megaprojects should use different evaluation indicator systems [2]. Project types are also set in different but specific contexts, and distinct social responsibility evaluation indicator systems for varied types of megaprojects of MMRI have not yet been established.

Fourth, the indicators from the perspective of social responsibility are more focused on the microlevel. However, current social responsibility research tends to focus only on the macrolevel results of national visions and strategic goals [42], with limited research on microlevels [25]. While current megaprojects purposes and national sustainable development strategy frameworks focus more broadly on national aspirations and strategic goals, challenges remain in translating these strategic goals to the microlevel [2]. These findings can hardly be translated into practical actions when each participant at the microlevel needs to be considered [25].

Furthermore, scholars have established indicator systems related to sustainability in different fields based on various methodologies [12,41,43–45]. The industrial ecology approach has been adopted in view of the environmental impacts of products throughout their life cycles [46,47]. Based on the principles of ecology, this approach is applied to achieve industrial ecology through ecological reorganization, ultimately obtaining multiple economic, social, and ecological benefits, and ultimately achieving the sustainable development of human society. In addition, other scholars have used other methods such as group decision-making [48], the LEED green building rating system [49–52], and lifecycle cost analysis [53,54]. In particular, multiple-criteria decision-making (MCDM) is gaining popularity as a comprehensive approach to ensure that all important aspects and interactions are considered [25,55–57]. The unique characteristics of different types of projects, the hierarchical nature of social responsibility, and the multifaceted nature of sustainability objectives should be considered and incorporated into the decision-making process. Based on the combination of fuzzy set theory and hierarchical analysis, the application of FAHP provides significant improvements in the solution of fuzzy and uncertain problems [58–61].

Therefore, this study developed a holistic indictor system based on fuzzy hierarchical analysis to provide a method to effectively evaluate MMRISR.

3. Methodology

This study applied various research methods to achieve the research objectives. The research process consisted of five main stages (Figure 1). We first selected the MMPRISR evaluation indicators by conducting a literature analysis of the factors affecting MMPRISR. Second, the indicators were screened on the basis of the initial indicators by means of a questionnaire-based survey and expert interviews. Then, FAHP was used to establish a model for the screened indicators and calculate the weights of each index layer and indicator layer. Finally, the case study was adopted to test the reliability of the indicator system.



Figure 1. Research framework and process.

3.1. Selection of the Primary Indicators

3.1.1. Social Responsibility Dimension Determination

The first stage for determining the appropriate and scientific set of MMRISR indicators is to select the initial indicators. Different models of social responsibility, such as the concentric circle model [38], pyramid model [31], and triple bottom-line model [62], have been reported in the literature. On the basis of those studies, social responsibility encompasses the economic, legal, social, environmental, philanthropic, and ethical aspects. One of the most widely used models is Carroll's social responsibility pyramid model [31]. In accordance with the classic pyramid model of social responsibility, this study adopted the social responsibility evaluation indicator system proposed by Lin et al. [2], based on MIPs, and subsequently added social responsibility indicators for depicting the unique background of MMRI. Finally, the social responsibility, legal responsibility, environmental and ethical responsibility, and political responsibility [2,14,27].

3.1.2. Identification of Indicator Sources

On the basis of the existing literature and the development of social responsibility research theories on MMRI, we reviewed many papers on the project management, social responsibility, and sustainable development of MIPs. The unprecedented worldwide attention to sustainable development and social responsibility has led several projects to aim for certifications showing their ability to conduct sustainable activities following international standards [2]. Therefore, in addition to the academic literature, this study considered international principles and guidelines as key sources of MSR indicators.

The sources of social responsibility indicators for MMRI are as follows [32]:

- Academic literature—A systematic search of relevant databases was conducted by using keywords such as "infrastructure project", "social responsibility", "stakeholder" and "indicator system", and the publication year is between 2001 and 2021;
- Relevant international standard systems for the social responsibility guidelines, such as ISO 26000;
- Relevant social responsibility reports related to MIPs, such as those issued by enterprises or organizations;
- (4) Relevant industry principles and guidelines, such as the Guidelines on Social Responsibility in the Chinese Foreign Contracting Engineering Industry; and
- (5) Relevant feasibility study reports, such as prefeasibility study decision reports for MMRI.

3.2. Screening for Indicator Optimization

The design of the questionnaires was based on the literature review, case studies, and expert interviews. Due to the limitations of the sample of megaprojects, the questionnaires used in this study were sent to 20 senior managers who had participated in the planning and establishment of MMRI to solicit an expert evaluation of social responsibility indicators at the stage of indicator determination and indicator relative importance judgment. Therefore, the established system of social responsibility evaluation indicators is only for the MMRI field. The specific content of questionnaire is given in Appendices A and B. Meanwhile, social responsibility indicators are highly subjective. Thus, the data sources in this study were strictly controlled during indicator selection. To maximize variability and generalizability, we sought out megaproject experts and senior managers of different sizes, located in different geographic locations, and demonstrating different levels of MSR performance. In particular, we ensured that the interviewed experts had participated in the construction of at least one MMRI. The experts are all highly educated, most of them have a master's degree or a higher educational attainment, and have intermediate or higher titles. The experts also include the following: those with government roles; owners; part of design, construction, supervision, and operation units; and those who had participated in the project establishment, design, construction and operation, and maintenance stages. Since the interviewed experts are senior managers with long experience in the engineering industry, they have more professional experience in this field to ensure the reliability of the survey, and the gender of all the experts is male, which basically conforms to the characteristics of gender imbalance in the engineering field. The background information on the interviewed experts is shown in Table 1. These data proportions basically reflect the actual situation of organizational forms in China. The data indicate that the interviewed experts have experience and knowledge of the issues related to this study, which adds to the credibility of the data. Furthermore, ten experts who had participated in the construction of MMRI were invited to judge the MMRISR evaluation indicator system in terms of the degree of recognition of the indicators.

3.3. Establishment of the Indicator System

In accordance with the MMRI perspective, four aspects of the MMRISR indicators were summarized from the existing studies. The MMRISR evaluation indicator system in this study was established by screening social responsibility indicators by means of questionnaires and expert interviews. The indicator system was stratified into three levels as follows. The first level is the target level, which represents the indicator system used to achieve the final purpose of the evaluation. The second level is the component level with four evaluation components: economic and quality responsibility, legal responsibility, environmental and ethical responsibility, and political responsibility. The third level is the factor level, in which each evaluation component can be determined on the basis of several

Classification Classification Type Percentage Type Percentage 100% Male 1 to 5 years 10% Gender 6 to 10 years Female 0 70% Years of work 21 to 30 years old 10% 10% 11 to 20 years 31 to 40 years old 70% More than 20 years 10% Age 41 to 50 years old 10% Government 10% >50 years old 10% Owners 10% Engineering Technology 50% Designers 50% Specialties Unit Roles Engineering Management 50% Constructor 20% Undergraduate 30% Supervisors 10% Academic Master 60% Operators 10% qualifications PhD 10% Project stage 70% Primary 10% Design Phase 70% Project Phase Intermediate 40%Construction Phase 60% Title Operation and Advanced 50% 10% maintenance phase

Table 1. Background information of interviewed experts.

3.4. Calculation of the Indicator Weights

FAHP is a multi-indicator evaluation method that combines the analytic hierarchy process (AHP) and the fuzzy comprehensive evaluation (FCE). Fuzzy hierarchical analysis, which is an extension of the basic hierarchical analysis for solving uncertain and ambiguous problems, is a multiple-criteria decision-making (MCDM) method for solving complex problems with uncertainties and ambiguity [64,65]. MMRI is a complex system. The evaluation indexes of MMRISR are similar to hierarchical relationships in that they involve more factors and have clear hierarchical relationships [2]. The majority of the MMRISR evaluation indicators are qualitative indicators, and the measurement language is fuzzy [66]. However, unlike economic evaluation indicators, the MMRISR indicators are difficult to quantify into specific numbers and formulas, and their language belongs to the domain of fuzzy boundaries [67,68]. Therefore, a numerical analysis was applied in the current research to calculate the weight priority of the MMRISR indicators, and FAHP was used for multi-criteria decision-making [69,70]. The steps involved in the weight calculation include the following:

evaluation factors. Finally, the evaluation indicator system was developed to include the

three levels (target, component, and factor levels) and screening indicators [63].

Step 1: Establish the fuzzy judgment matrices.

In fuzzy hierarchical analysis, the fuzzy judgment matrix can be derived by instructing experts to compare indicators according two parametric importance. The two-by-two comparison judgment between the factors is expressed quantitatively using the importance of one factor over another factor. The fuzzy judgment matrix is given by $A = (A_{ij})_{n*n'}$, where A is the fuzzy complementary judgment matrix, and a_{ij} is the degree of affiliation (i.e., "more important than"). The magnitude of this degree is denoted by a fuzzy judgment scale with values ranging from 0.1 to 0.9. From the importance linguistic scale, the factors can then be compared with each other, and the following fuzzy complementary judgment matrix is obtained:

	<i>a</i> ₁₁	a_{12}	•••	<i>a</i> _{1n}
A =	a ₂₁	a ₂₂	•••	<i>a</i> _{2<i>n</i>}
		•••	•••	
	a_{n1}	a_{n2}	•••	a_{nn}

Step 2: Calculate the relative importance of the factors.

In general, FAHP requires multiple experts to perform the evaluation, which can be denoted as:

$$A_i^{(l)} = \sum_{k=1}^n a_{ik}^{(l)}, i = 1, 2, \dots, n, l = 1, 2, \dots, s$$
⁽¹⁾

In real decision-making, the obtained fuzzy judgment matrix is contradictory, due to the complexity of parameters and the one-sidedness of understanding these parameters. Hence, knowing the consistency of a fuzzy judgment matrix is essential. According to the definition of the fuzzy consistency matrix, the mathematical transformation is performed as follows:

$$b_{ij}^{(l)} = \frac{a_i^{(l)} - a_j^{(l)}}{2(n-1)} + 0.5, l = 1, 2, \dots, s.$$
⁽²⁾

The fuzzy consistency matrix is obtained as follows:

$$B = \left(b_{ij}^{(l)}\right)_{n \times n'} (l = 1, 2, \dots, s).$$
(3)

Suppose the number of experts is *m*, and each expert is given the same weight of $\lambda_1 = \lambda_2 = \dots \lambda_{\infty} = \delta$. Then,

$$\omega_i = \frac{\sum_{i=1}^n \sum_{j=1}^n \lambda b_{ij}^{(l)} + \frac{n}{2} - 1}{n(n-1)}$$
(4)

The relative weights of each indicator in the matrix can be obtained by ω_i .

Step 3: Rank the overall importance of each factor.

The relative importance of the low-level factors with respect to the high-level factors are calculated. Then, the importance is ranked on the basis of the above steps to determine the position of each factor in the overall evaluation index system.

3.5. Comprehensive Evaluation

Step 1: Establish the single-factor fuzzy judgment matrix.

Establish a single-factor evaluation rubric set $V = \{V1, V2, ..., Vm\}$, for the qualitative indicators, and let the experts involved in the evaluation rate the indicators U_i . Count the frequency of evaluation results belonging to rank V_iM_{ij} . Calculate:

$$r_{v_j}(D_i) = \frac{M_{ij}}{N} \tag{5}$$

where M_{ij} is the number of evaluation results belonging to V_i (number of times), and N is the number of experts involved in the evaluation. $r_{v_i}(D_i)$ is $D \in V_i$ (affiliation degree).

Then, the indicator U_i (affiliation function) is calculated as:

$$r_{v_j} = \frac{r_{v_1}(D_i)}{V_1} + \frac{r_{v_2}(D_i)}{V_2} + \dots + \frac{r_{v_m}(D_i)}{V_m}.$$
(6)

By collecting, organizing, and counting the questionnaires, the one-factor judgment matrix (R of U to V) can be established as:

$$R = \begin{bmatrix} r_{11} & r_{12} & \cdots & r_{1m} \\ r_{21} & r_{22} & \cdots & r_{2m} \\ \vdots & \vdots & \ddots & \vdots \\ r_{n1} & r_{n2} & \cdots & r_{nm} \end{bmatrix}$$

Step 2: Establish the comprehensive evaluation model.

$$r = W \cdot R = (w_1, w_2, \dots, w_n) \begin{bmatrix} r_{11} & r_{12} & \cdots & r_{1m} \\ r_{21} & r_{22} & \cdots & r_{2m} \\ \vdots & \vdots & \ddots & \vdots \\ r_{n1} & r_{n2} & \cdots & r_{nm} \end{bmatrix} = (r_1, r_2, \dots, r_m)$$
(7)

The fuzzy synthetic operation in Formula (7) is the ordinary matrix multiplication method (weighted average method). This model allows each factor to contribute to the comprehensive evaluation, objectively reflecting the entire status of the evaluation object.

4. Analysis and Results

4.1. Analysis of Indicator Screening

From the questionnaire survey, 20 experts were allowed to score 32 social responsibility indicators. The normalized and ranked results are shown in Table 2. The indicators whose normalized values were equal to or higher than 0.4 were selected as the key measures of the MMRISR evaluation.

The rationality of the indicator system was also confirmed with the experts, and the reasons for excluding the indicators with normalized values lower than 0.4 were analyzed. The responses included the following explanations. First, the indicator already has a clear responsible subject, and the relevant parties are already relatively good in certain aspects (e.g., compensation for land acquisition and demolition, reasonable use of construction resources, active organization of public participation, technological innovation and application, and road maintenance). Second, other indicators (project reporting compliance, project reporting independence and impartiality, and public opinion supervision) are less important than the other indicators. Therefore, 24 indicators for social responsibility evaluation were screened for the establishment of the FAHP model.

Number	Indicators	Average Value	Normalization	Ranking
1	Meets the needs of residents' life and travel	4.9	1.00	1
2	Public event emergency response	4.9	1.00	1
3	Quality and safe construction of road projects	4.9	1.00	1
4	Water, noise, dust, and other pollution control	4.8	0.94	4
5	Regional transportation network construction	4.7	0.88	5
6	Proper handling of the relationship between people, vehicles, roads, and the environment	4.7	0.88	5
7	Project technical feasibility decision	4.7	0.88	5
8	Raising awareness of environmental protection	4.7	0.88	5
9	Road construction cost and schedule control	4.6	0.82	9
10	Design, construction, and operation comply with transportation industry specifications and legal requirements	4.6	0.82	9
11	Environmental ecological and cultural protection along the road	4.6	0.82	9
12	Consideration of the impact of the proposed road on existing tracks	4.5	0.76	12
13	Protecting the rights and interests of participating employees	4.5	0.76	12
14	Coordinate the sequence of road construction and pipeline construction	4.5	0.76	12
15	Sound engineering project governance mechanism	4.4	0.71	15
16	Interoperability with the surrounding environment	4.4	0.71	15
17	Monitoring and reporting wrongdoing	4.4	0.71	15
18	Maintaining relationships with the surrounding community	4.3	0.65	18
19	Project economic feasibility decision	4.3	0.65	18
20	Project operating costs and safety assurance	4.3	0.65	18
21	Effective coordination between the main government departments	4.2	0.59	21
22	Information disclosure	4.2	0.59	21

Table 2. Ranking of MMRISR indicators.

Number	Indicators	Average Value	Normalization	Ranking
23	Engineering anticorruption	4	0.47	23
24	Effective regulation	4	0.47	23
25	Rational use of construction resources	3.8	0.35	25
26	Road maintenance	3.8	0.35	25
27	Project coverage compliance	3.8	0.35	25
28	Technology innovation and application	3.8	0.35	25
29	Project reporting independence and impartiality	3.8	0.35	25
30	Compensation for land acquisition and relocation	3.5	0.18	30
31	Public opinion monitoring	3.3	0.06	31
32	Actively organizes public participation	3.2	0.00	32

Table 2. Cont.

4.2. Establishment of the Indicator System

The MMRISR evaluation indicator system was divided into three levels: target, component, and factor levels. Social responsibility was divided into four dimensions: economic and quality, legal, environmental and ethical, and political. After the screening, 24 indicators were retained to establish the MMRISR evaluation indicator system. Figure 2 shows the hierarchy model of the MMRISR evaluation indicators.

4.3. Analysis of Indicator Weighting

The questionnaire data were used for the weighting. The fuzzy judgment matrices of the first-level social indicators A and second-level social responsibility indicators U_1 , U_2 , U_3 , and U_4 were obtained using Formula (1).

$$A = \begin{bmatrix} 0.5 & 0.59 & 0.58 & 0.57 \\ 0.41 & 0.5 & 0.57 & 0.58 \\ 0.42 & 0.43 & 0.5 & 0.6 \\ 0.43 & 0.42 & 0.4 & 0.5 \end{bmatrix}$$
$$U_{1} = \begin{bmatrix} 0.5 & 0.43 & 0.63 & 0.68 & 0.65 & 0.63 & 0.54 \\ 0.57 & 0.5 & 0.62 & 0.63 & 0.65 & 0.57 & 0.55 \\ 0.37 & 0.38 & 0.5 & 0.58 & 0.55 & 0.54 & 0.52 \\ 0.32 & 0.37 & 0.42 & 0.5 & 0.55 & 0.53 & 0.51 \\ 0.35 & 0.35 & 0.45 & 0.45 & 0.5 & 0.58 & 0.51 \\ 0.37 & 0.43 & 0.46 & 0.47 & 0.42 & 0.5 & 0.58 \\ 0.46 & 0.45 & 048 & 0.49 & 0.49 & 0.42 & 0.5 \end{bmatrix}$$
$$U_{2} = \begin{bmatrix} 0.5 & 0.45 & 0.63 & 0.47 & 0.53 & 0.53 \\ 0.55 & 0.5 & 0.58 & 0.51 & 0.59 & 0.53 \\ 0.37 & 0.42 & 0.5 & 0.51 & 0.56 & 0.5 \\ 0.53 & 0.49 & 0.49 & 0.5 & 0.59 & 0.58 \\ 0.47 & 0.41 & 0.44 & 0.41 & 0.5 & 0.54 \\ 0.47 & 0.47 & 0.5 & 0.42 & 0.46 & 0.5 \end{bmatrix}$$



Figure 2. Hierarchy model of the MMRISR evaluation indicators.

The fuzzy mathematical transformation was conducted according to Formula (2), and the fuzzy consistent judgment matrix of M_1 , N_1 , N_2 , N_3 , and N_4 was obtained using Formula (3). Consider that the fuzzy judgment matrix A is composed of level-1 social responsibility indicators. The weight calculation process is as follows.

The mathematical transformation is performed using the formula of fuzzy judgment matrix A to obtain the fuzzy consistency matrix M_1 .

	0.5	0.53	0.55	0.58
N/	0.47	0.5	0.52	0.55
$\mathbf{M}_1 =$	0.45	0.48	0.5	0.53
	0.42	0.45	0.47	0.5

The weights subsequently are calculated using Formula (4), i.e., $\omega_1 = (0.2633, 0.2533, 0.2472, 0.2361)^T$.

For the weight coefficients of each evaluation indicator at the criterion level, the following results were obtained: political responsibility (0.2633), economic and quality responsibility (0.2533), legal responsibility (0.2472), and environmental and ethical responsibility (0.2361). Similarly, the fuzzy judgment matrix within the criterion and the relative importance of each index was derived from the questionnaire data. The final results are shown in Table 3.

The findings indicate that the MMRI should comprehensively consider social responsibility with respect to the political, economic and quality, legal, and environmental and ethical aspects.

For political responsibility, meeting the living and travel needs of residents and building a regional transportation network obtained the top two ranks. This finding reflects the purpose of the MMRI, which is to provide services to the public as a means of improving the quality of life of residents [71]. In this regard, the government plays a key role in social responsibility management because it is the subject of political responsibility. Furthermore, the government has the final decision-making power on the construction of MMRI [72]. If the government's exercise of public power ignores regulation and restraint, then it will cause significant irreparable damage to social stability, national property, and people's life safety. The implementation of anticorruption measures is also important in avoiding potential quality problems in engineering and safety accidents [73]. In addition, the government's emergency response to public events also serves as a guarantee for maintaining social stability.

For economic and quality responsibility, the indicators of project technical feasibility and road project quality and safety construction ranked the highest. Technical decisionmaking, quality control, and safe construction represent the basic objectives of MMRI [9]. Technical errors in preproject decision-making or quality defects during construction will lead to construction accidents or human casualties. In addition, as MMRI is a public good, it is a nonprofit project invested in by the government to improve the people's accessibility and accelerate urban development [10]. However, given the financial restrictions of the government, scientific and reasonable economic decisions can be implemented in the long run to support the construction of non-operating MIPs, achieve the sustainable development of MIPs, and fulfill the sustainability requirements of the MMRISR.

For legal responsibility, it includes the code of conduct of relevant participants with respect to the design, construction, and operation norms in the transportation industry and the related legal requirements [14]. Breaches in legal defense indicate many types of opportunistic behavior [74]. Promoting information disclosure and effective supervision of MMRI is a good approach to realizing the governance of MMRISR.

For environmental and ethical responsibility, strengthening the pollution control of water, noise, and dust and raising awareness of environmental protection are the most popular topics at present [18,75]; in China, they correspond to the increasing emphasis on the concept of sustainable development [76,77]. Only by respecting nature and protecting the environment can the people and society achieve sustainable development [78].

Target Level	Component Level	Factor Level	Combined Weights
		Regional transportation network construction (0.1506)	0.03966
		Meets the living and travel needs of residents (0.1522)	0.03977
		Maintaining neighboring community relations (0.1420)	0.03739
	Political responsibility (0.2633)	Effective coordination among governmental subject departments (0.1387)	0.03652
		Engineering anticorruption (0.1386)	0.03648
		Proper handling of the relationship between people, vehicles, roads, and the environment (0.1391)	0.03663
		Public event emergency response (0.1399)	0.03685
		Project economic feasibility decision (0.1689)	0.04277
		Project technical feasibility decision (0.1719)	0.04353
	Economic and quality responsibility (0.2533)	Well-established governance mechanism	0.04151
Social responsibility evaluation of major municipal road infrastructure		responsibility (0.2533) Quality and safe construction of road works (0.1701)	
		Road construction cost and schedule control (0.1621)	0.04105
		Project operating costs and safety assurance (0.1631)	0.04130
		Information disclosure (0.3508)	0.08673
	Legal responsibility	Effective regulation (0.3271)	0.08086
	(0.2472)	accordance with transportation industry specifications and legal requirements (0.3221)	0.07962
		Water, noise, dust and other pollution control (0.1271)	0.03002
		Environmental eco-cultural protection along the road (0.1262)	0.02980
		Consideration of the impact of the proposed road on existing tracks (0.1251)	0.02954
	Environmental and ethical	Interoperability with the surrounding environment (0.1248)	0.02946
	responsibility (0.2361)	Safeguarding the rights and interests of participating employees (0.1261)	0.02978
		Coordinate road construction with pipeline construction sequence (0.1224)	0.02891
		Raising awareness of environmental protection (0.1269)	0.02997
		Supervision and prosecution of wrongdoing (0.1212)	0.02862

Table 3. Weight results for MMRISR indicator system.

5. Case Study

5.1. Case Background

The Zhengzhou 107 Auxiliary Road Project is located in Zhengzhou City, Henan Province, China. The auxiliary road is an eight-lane road in both north and south directions. Its total length is approximately 20 km and comprises six interchanges. The auxiliary road project is one of the megaprojects classified as a MIP. The Zhengzhou 107 Auxiliary Road Project is a fast-running lane and has been put into operation. A social responsibility evaluation of this project will allow for subsequent megaprojects to be sustainable.

5.2. Acquisition and Quantification of the Evaluation Indicator Data

All evaluation results should be accurate and reliable. In this study, questionnaires were issued to design and construction experts who had participated in constructing the Zhengzhou 107 Auxiliary Road Project. Ten valid questionnaires were received. The questionnaire entailed a five-level rubric set in which $V = \{excellent, good, fair, poor, very poor\} = \{V1, V2, V3, V4, V5\}$. The single-factor evaluation was conducted in the questionnaire format to derive the rubric for each factor. The results of the social responsibility evaluation of the Zhengzhou 107 Auxiliary Road Project were obtained by collecting, organizing, and counting the questionnaires.

5.3. Application of the Evaluation Model

The affiliation degree of each index is calculated using Formula (6). The results are shown in Table 4.

	In diastana		Eva	luation Sta	atus	
	Indicators	Excellent	Good	General	Poor	Very Poor
	Regional transportation network construction	0.8	0.1	0.1	0	0
	Meets the needs of residents' life and travel	0.9	0.1	0	0	0
	Maintaining relationships with the surrounding community	0.2	0.7	0.1	0	0
Political responsibility	Effective coordination between the main government departments	0.3	0.4	0.3	0	0
	Engineering anticorruption	0.2	0.4	0.4	0	0
	Proper handling of the relationship between people, vehicles, roads, and the environment	0.2	0.2	0.2	0.3	0.1
	Public event emergency response	0.1	0.3	0.6	0	0
	Project economic feasibility decision	0.1	0.8	0.1	0	0
Economic and	Project technical feasibility decision	0.3	0.2	0.3	0.2	0
Quality	Sound engineering project governance mechanism	0.1	0.4	0.4	0.1	0
Responsibility	Quality and safe construction of road projects	0.5	0.1	0.4	0	0
Responsibility	Road construction cost and schedule control	0.1	0.4	0.4	0.1	0
	Project operating costs and safety assurance	0.1	0.5	0.4	0	0
	Information disclosure	0.2	0.1	0.7	0	0
	Effective regulation	0.2	0.8	0	0	0
Legal Liability	Design, construction, and operation comply with					
	transportation industry specifications and	0.2	0.6	0.2	0	0
	legal requirements					
	Water, noise, dust, and other pollution control	0.3	0.3	0.4	0	0
	Environmental ecological and cultural protection along the road	0.2	0.7	0.1	0	0
Environmental	Consideration of the impact of the proposed road on existing tracks	0.2	0.7	0.1	0	0
and Ethical	Interoperability with the surrounding environment	0.3	0.1	0.3	0.2	0.1
Responsibility	Protecting the rights and interests of participating employees	0.2	0.3	0.3	0.2	0
	Coordinate the sequence of road construction and pipeline construction	0.2	0.2	0.4	0	0.2
	Raising awareness of environmental protection	0.1	0.4	0.3	0.2	0
	Monitoring and reporting wrongdoing	0.2	0.3	0.4	0.1	0

Table 4. MMRISR evaluation indicator survey statistics.

The single-factor fuzzy judgment matrix (R from U to V) can be established using the index affiliation table. The comprehensive evaluation results of the model are calculated using Formula (7), where W is the weighted weight of the social responsibility evaluation indexes obtained via FAHP.

	0.03966 -	0.8	0.1	0.1	0	0 -	
	0.03977	0.9	0.1	0	0	0	
	0.03739	0.2	0.7	0.1	0	0	
	0.03652	0.5	0.4	0.1	0	0	
	0.03648	0.2	0.4	0.4	0	0	
	0.03663	0.2	0.2	0.2	0.3	0.1	
	0.03685	0.1	0.3	0.6	0	0	
	0.04277	0.1	0.8	0.1	0	0	
	0.04353	0.3	0.2	0.3	0.2	0	
	0.04151	0.1	0.4	0.4	0.1	0	
	0.04308	0.5	0.1	0.4	0	0	
F = W R =	0.04105	0.1	0.4	0.4	0.1	0	-(0.26, 0.40, 0.28, 0.05, 0.01)
$L = W \cdot K =$	0.04130	0.1	0.5	0.4	0	0	$\left[\begin{array}{c} -(0.20, 0.40, 0.23, 0.05, 0.01) \\ \end{array}\right]$
	0.08673	0.2	0.1	0.7	0	0	
	0.08086	0.2	0.8	0	0	0	
	0.07962	0.2	0.6	0.2	0	0	
	0.03002	0.3	0.3	0.4	0	0	
	0.02980	0.2	0.7	0.1	0	0	
	0.02954	0.2	0.7	0.1	0	0	
	0.02946	0.3	0.1	0.3	0.2	0.1	
	0.02978	0.2	0.3	0.3	0.2	0	
	0.02891	0.2	0.2	0.4	0	0.2	
	0.02997	0.1	0.4	0.3	0.2	0	
	0.02862	0.2	0.3	0.4	0.1	0	

5.4. Analysis of the Evaluation Results

The obtained FCE result of the affiliation vector E suggests a certain gap in the affiliation values of each result level; the maximum affiliation principle can be used to discriminate the values. According to the principle of maximum affiliation discrimination, the result level with a large affiliation is determined as the final evaluation level. The maximum value is 0.40; therefore, the MMRI final evaluation result is 0.40. The results of the MMRISR evaluation are divided into four grades. As shown in Table 5, the MMRISR evaluation result of the Zhengzhou 107 Auxiliary Road Project is "good," implying that the governance of the MSR can be improved.

Table 5. The rank of evaluation results.

Evaluation results	0–0.1	0.1–0.2	0.2–0.3	0.3–0.4	0.4–0.5
Rank	Very poor	Poor	General	Good	Excellent

In terms of political responsibility, although the planning and establishment of the project can fully build a regional transportation network and meet the needs of public life and travel, shortcomings are apparent. First, the project managers did not conduct a comprehensive planning before implementation, which led to the increase in various costs during the construction period. Second, the road planning was not perfect, and the relation of people, vehicles, roads, and the environment was not properly handled. A secondary design was consequently implemented during the construction of the project, which led to the delay of the construction period. Third, the organization and coordination of government departments is poor. For example, for the railroad-crossing sections and military

facilities along the route, various departments overlooked their responsibilities, resulting in a huge waste of resources. Problems related to economic and quality responsibility were also observed during project implementation. Oftentimes, quality issues are concealed by economic benefits, which is rooted in the orientation of maximizing all benefits [74]. For example, the stakeholders only consider the direct cost of the project for the cost-saving measures, whereas the indirect cost and social cost are ignored. Furthermore, the contractor used "jerry-built materials" for profit, leading to serious quality problems. Regarding the legal responsibility aspect, all parties involved in the construction can restrain one's own behavior according to the legal specifications and design requirements. However, the accountability mechanism of the project is inappropriate. As delineations of responsibilities are lacking, the penalties pertaining to the liability problems are often weak to create a warning effect [27]. The environmental aspects of the project were controlled by dust control during implementation, which greatly improved the environmental pollution problem as opposed to the schemes of previous projects [14]. On the one hand, the measures are mandatory and ensured by the environmental protection department of the government. On the other hand, the project participants seldom took the initiative to fulfill their environmental responsibilities [15], and their awareness remained at a low level.

The aforementioned problems reveal certain shortcomings in MMRISR. By further reforming and improving the decision-making mechanism, the negative effects related to the lack of social responsibility can be greatly improved—for example, by consulting with experts and people from all walks of life, increasing the transparency of decision-making, and raising the awareness of social responsibility among citizens. Undeniably, the project construction has improved the mobility and life of the residents, helped to build a good relationship with the community, and increased the resilience of the MIPs. The positive benefits of the project will more than compensate for the damage to the surrounding environment and livelihoods, ultimately achieving the goal of sustainable development.

The MMRISR evaluation result of Zhengzhou 107 Auxiliary Road Project is generally consistent with the project situation learned through the expert interviews, thus verifying the reliability and usefulness of the proposed MMRISR indicator system.

6. Discussion and Implication

6.1. Discussion

To date, an adequate and comprehensive MMRISR evaluation indicator system does not exist. The system could have addressed MMRISR performance in a comprehensive and detailed manner and translated the strategic objectives into practical actions. Our research has proposed an MMRISR indicator system that builds upon the existing indicator framework for a megaproject's social responsibility by developing a conceptual framework embedded in MMRI. This study focuses on the social responsibility evaluation of major municipal road infrastructure, and the theoretical construction and empirical results verify the reliability and practicality of the system.

In terms of the four dimensions, the experts believe that the core of the evaluation of MMRISR is "political responsibility". MMRI not only assumes the role of regional network building in the political aspect but can also maintain the surrounding community's relations and fight corruption in engineering projects. The main body of political responsibility relies on the government. Although the four dimensions have similar weight values, the weight of political responsibility is higher than those of the other elements. The analytic results showed that political responsibility represents the highest priority in China's unique government–market institutional context [15]. Given the political connections, major engineering infrastructure projects are usually recognized as a measure of political performance [74], further indicating that political responsibility requires more emphasis. Through the successful completion of MIPs, the participants refer to their desire for political promotions, political access, political appointments, higher levels of government support, and connections to the government [79,80]. As a result, it is possible to improve information sharing and communication between MIPs participants and the government,

gain access to potential market opportunities, and help them to benefit from incentives and subsidies in the future [81].

In terms of the 24 indicators, among the top factors, "information disclosure" is necessary for evaluating MMRISR. Meanwhile, "effective regulation" and "design, construction, and operation in accordance with transportation industry specifications and legal requirements," belonging to the legal responsibility classification, are also imperative for evaluating MMRISR. This result suggests that legal responsibility can be fulfilled through externalities, such as public and media supervision [79,82,83]. The participants could restrain their behavior by focusing on the legal norms and design requirements, but the accountability mechanism is generally faulty [14]. The phenomenon of unclear division of responsibility has caused parallel management and multiple management, leading to the emergence of liability problems caused by the loss of either an equal share of responsibility or the law. Subsequently, unlawful practice is difficult to punish, and the problem at the onset could not be solved. Meanwhile, the public is increasingly concerned about its impact on health and well-being [84]. Therefore, design, construction, and operation should not only comply with industry norms and legal requirements but also achieve effective public and media supervision, information disclosure, and public opinion monitoring, as these indicators are essential for the fulfillment of MMRISR [85]. For example, deficiencies in social responsibility can be reduced by regularly publishing social responsibility reports. In addition, improve the relevant laws. These measures are very important to promote the MMRISR fulfillment. The public and media can help to effectively monitor certain concerns [86,87]. Most of the major engineering social responsibility infrastructure projects remain weak in terms of awareness of legal responsibility. The research results show that information disclosure, effective supervision, and compliance with legal requirements to improve regulation of MMRISR should be the focus of MMRISR fulfillment. Furthermore, MMRI should be built under strict supervision and information disclosure, adhering to the concept of social responsibility and sustainability.

Economic and quality responsibility is also necessary in MMRISR evaluation. The weights of all indicators pertaining to economic and quality responsibility are also relatively high. This scenario indicates that the iron triangle (cost, time, and scope) of traditional engineering project management remains to be a key indicator of preproject decision-making and engineering performance measurement [9,88–90], as cost overruns on MMRI projects are common [88,89,91,92]. Furthermore, the technical feasibility decision at the project stage and the quality and safety construction of road projects are a common concern. If the lack of social responsibility in the aspects of economy and quality leads to quality-related accidents, then waste of resources and loss of social welfare will likely occur [14]. In terms of environment and ethics, pollution control and environmental protection have been taken seriously by all parties participating in the construction [18,77], but the awareness of ethical responsibilities, such as employee rights protection and monitoring and reporting of negligence, fraud, bribery, and dishonest wrongdoings, is relatively weak [78,93,94].

MSR research is being conducted worldwide and has been gaining more and more attention from scholars as it plays an important role in sustainable development. However, due to the different conditions of each country, each country needs to establish an MM-RISR evaluation index system according to its own specific needs. Based on the unique government–market dichotomy political context in China, megaprojects are not only public infrastructure, but also a political task. Therefore, the MMRISR evaluation indictor system must include meeting residents' living and travel needs, maintaining surrounding community relations, and properly handling the relationships between people, vehicles, roads, and the environment. In addition, the MMRISR development evaluation indictor system places more emphasis on simple implementation and effective evaluation, and is easy to promote in decision-making and construction units. For example, assessing the quality of road works and road construction cost system are practical measures.

The experts and senior managers have fully considered the evaluation indicators, thus ensuring the reliability and applicability of the evaluation results. The MMRISR

indicator system provides a useful reference for future MMRI construction. Instead of just pursuing fast construction, high yield and low cost, we should focus on regional economic development, ecological protection, community stability, and sustainable development. According to the actual situation in China, policy makers should fully consider the enforceability of policies as well as substantial incentives, such as favorable tax policies, loan support, and more project support from the government, in order to stimulate the willingness to actively fulfill MMRISR. This study provides an in-depth analysis of the experts' assessments. By summarizing the existing literature, preliminary evaluation indicators were screened: then, based on the characteristics and needs of MMRI projects, an MMRISR evaluation indicator system suitable for China was identified through FAHP, which provides a useful reference for promoting sustainable development strategies for major municipal infrastructure in China.

6.2. Management Implication

This study entails management implications for project decisionmakers in MIP management. First, the feasibility study report, which is often used as the basis for the construction project, is a key document in the early stages. However, social responsibility as a component of the feasibility study often involves only a short qualitative description of social responsibility; it neither draws attention to stakeholders nor achieves the real evaluation purpose of social responsibility [95,96]. In this study, we recommend improvements to the feasibility study evaluation mechanism by including a social responsibility evaluation in the prefeasibility study report to fundamentally strengthen the social responsibility discussion.

Second, considering the positive role of MMRISR, stakeholders of MIPs are encouraged to strive for MMRISR in practice to achieve the sustainable development of MIPs through economic and quality, legal, and political issues [2]. Third, due to the institutional deficiencies in the industry's social responsibility, the relevant laws and regulations are lacking, further indicating that correcting social responsibility deficiencies is weak. Strong legal support for MMRISR management can be enhanced by developing and improving effective policy tools and crafting new laws and rules related to social responsibility. This initiative can help to solve the problem of MIPs' weak sustainable development [27,97].

7. Conclusions and Future Research

7.1. Conclusions

Megaproject sustainability encompasses key aspects of social responsibility, environmental protection, and economic profitability. As a major public infrastructure project, MMPI will have far-reaching effects on the politics, economy, society, environment, and public of the region. Therefore, social responsibility, as an important practice for sustainable development of megaprojects, currently plays an essential role in improving environmental conditions and maintaining social stability. Our findings showed that the lack of social responsibility is closely related to the lack of attention to project evaluation and the unclear social responsibility in the predecision stage. This study introduces the concept of social responsibility in MMPI, screens social responsibility indicators for MMRI, and develops an MMRISR evaluation indicator system to provide a new perspective for MMRI governance.

The MMRISR indicator system proposed in this study was established via the methods of literature review, questionnaire survey, and FAPH. First, this study determined the dimensions and primary indicators through a literature review. Second, the MMRISR evaluation indicator system was established by screening 24 social responsibility indicators for MMRI. A questionnaire survey and expert interviews were conducted, covering the four dimensions of economic and quality responsibility, legal responsibility, environmental and ethical responsibility, and political responsibility. Then, FAHP was used to calculate the weights of the social responsibility indicators. Finally, the findings were applied to the Zhengzhou 107 Auxiliary Road Project as the case study to prove the feasibility of the model's application to a practical example.

The research contributions of this study are as follows. First, this study has established the MMRISR evaluation indicator system. This indicator system not only helps to improve the science basis of project establishment and decision-making but also strengthens the attention and implementation of social responsibility of key stakeholders. Moreover, the method and process of establishing the indicator system can provide reference for the establishment of MSR evaluation criteria for other types of projects. Second, this study has developed a weighted indicator system. This approach can overcome the obstacles of quantifying social responsibility issues. The weighting priority can reflect the problems and trends of MMRISR management, and it further provides directions for project decision makers for developing effective systems or policies. Third, with a strong theoretical basis, the indicator system can be subsequently incorporated into regulatory systems (e.g., feasibility studies) and provide a decision-making tool for MMRISR governance. When the social responsibility of implemented projects are evaluated, we can grasp whether the fulfillment of social responsibility deviates from the social responsibility requirements in the prefeasibility study and correct deviations on time to fundamentally promote the sustainable development of MMRI. Finally, this study enriches the theory of social responsibility and sustainability of major projects, and it is conducive to the promotion of sustainable development of MIPs. Similarly, the findings offer a practical value for guiding the government to effectively govern MSR.

7.2. Limitations and Future Research

This study has several limitations, but it also reveals promising areas for future research. First, the MMRISR evaluation indicator system was established on the basis of unique Chinese conditions. Differences in cultural background and economic development imply a variety of social responsibility measures for adoption by different countries and regions. Whether the MMRISR evaluation indicator system is applicable to engineering projects in general, or to other countries around the world, remains to be verified by future studies. Second, the complex impact mechanism of social responsibility in MMRI and the numerous stakeholders both require proper attention. In this research, we only studied the social responsibility that should be assumed at the project level. We did not study in depth the interaction and transmission mechanisms among the socially responsible subjects and the social responsibility governance strategies. These topics are worthy of further research.

Author Contributions: Conceptualization, D.Y. and J.L.; methodology, J.L. and D.Y.; validation, J.P.; formal analysis, J.L.; investigation, D.Y. and J.P.; resources, L.L. and J.Z.; data curation, J.L. and L.L.; writing—original draft preparation, J.L.; writing—review and editing, J.P. and J.Z.; funding acquisition, D.Y. and J.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the National Natural Science Foundation of China (Grant-Nos.71801083; 71901220; 71871096; 71901113), Special major projects for research and development of Henan Province (Scientific and technological projects) (GrantNo.212102310002); Graduate Key Project for College of Engineering Management and Real Estate of Henan University of Economics and Law (Grant No.2021-2).

Conflicts of Interest: The authors declare no conflict of interest.

Target Level	Component Level	Level of I	Endorsement (1 Necessarily; 4	Strongly Di Agree; 5 Str	sagree; 2 Disa congly Agree)	gree; 3 Not
	Compensation for land acquisition and relocation	1	2	3	4	5
	Regional transportation	1	2	3	4	5
	network construction Meets the needs of residents' life and travel	1	2	3	4	5
	Maintaining relationships with the	1	-	3	-	5
Political	surrounding community	1	2	3	4	5
responsibility	Effective coordination between the main	1	2	3	4	5
	Engineering anticorruption	1	2	3	4	5
	Proper handling of the relationship					
	between people, vehicles, roads and	1	2	3	4	5
	Public event emergency response	1	2	3	4	5
	Supplementary			-		-
	Project economic feasibility decision	1	2	3	4	5
	Project technical feasibility decision	1	2	3	4	5
	Technology innovation and application	1	2	3	4	5
Financial and Legal Liability Proje	Sound engineering project	1	2	3	4	5
	governance mechanism					
	road projects	1	2	3	4	5
	Road construction cost and	1	2	2	4	-
	schedule control	1	2	3	4	5
	Road maintenance	1	2	3	4	5
	Project operating costs and safety assurance	1	2	3	4	5
	Supplementary					
	Information disclosure	1	2	3	4	5
	Effective regulation	1	2	3	4	5
	Design, construction, and operation comply	1	2	3	4	5
T 1 T 1. 11 (and legal requirements	1	2	3	4	5
Legal Liability	Project coverage compliance	1	2	3	4	5
	Project reporting independence	-	2	0	-	-
	and impartiality	1	2	3	4	5
	Supplementary					
	Water, noise, dust, and other	1	2	3	4	5
	pollution control	-	-	0	1	U
	Environmental ecological and cultural	1	2	3	4	5
	Consideration of the impact of the					
	proposed road on existing tracks					
	Interoperability with the				_	_
Environmental	surrounding environment	1	2	3	4	5
and Ethical	Protecting the rights and interests of	1	2	3	4	5
Responsibility	participating employees	1	2	5	-	5
	Rational use of construction resources	1	2	3	4	5
	Coordinate the sequence of road	1	2	3	4	5
	Public opinion monitoring	1	2	3	4	5
	Raising awareness of	1	<u>ـ</u>	5	- T	5
	environmental protection	1	2	3	4	5
	Monitoring and reporting wrongdoing	1	2	3	4	5
	Supplementary					

Appendix A. Questionnaire on MMRISR Indicators

Appendix B. Questionnaire on the Relative Importance of MMRISR Indicators

1. At the level of social responsibility indicators for large municipal road projects, please compare the relative importance of political responsibility U_1 , economic and quality responsibility U_2 , legal responsibility U_3 , environmental and ethical responsibility U_4 .

	Extremely Important	Very Important	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{1}/U_{2}									
U_1/U_3									
U_1/U_4									
U_{2}/U_{3}									
U_{2}/U_{4}									
U_{3}/U_{4}									

2. At the level of political responsibility (*U*) for large municipal road projects (1), please compare the relative importance of the following secondary indicators two by two. Construction of regional transportation network, U_{11} ; meeting residents' living and travel needs, U_{12} ; maintaining relations with surrounding communities, U_{13} ; effective coordination among main government departments, U_{14} ; engineering anticorruption, U_{15} ; proper handling of relations among people, vehicles, roads and environment, U_{16} ; emergency response to public events, U_{17} .

	Extremely Important	Very Important	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{11}/U_{12}									
U_{11}/U_{13}									
U_{11}/U_{14}									
U_{11}/U_{15}									
U_{11}/U_{16}									
U_{11}/U_{17}									
U_{12}/U_{13}									
U_{12}/U_{14}									
U_{12}/U_{15}									
U_{12}/U_{16}									
U_{12}/U_{17}									
U_{13}/U_{14}									
U_{13}/U_{15}									
U_{13}/U_{16}									
U_{13}/U_{17}									
U_{14}/U_{15}									
U_{14}/U_{16}									
U_{14}/U_{17}									
U_{15}/U_{16}									
U_{15}/U_{17}									
U_{16}/U_{17}									

3. At the level of economic and quality responsibility (U_2) for large municipal road projects, you are invited to compare the relative importance of the following secondary indicators two by two. Project economic feasibility decision, U_{21} ; project technical feasibility decision, U_{22} ; perfect project governance mechanism, U_{23} ; road project

	Extremely Important	Very IM- PORTANT	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{21}/U_{22}									
U_{21}/U_{23}									
U_{21}/U_{24}									
U_{21}/U_{25}									
U_{21}/U_{26}									
U_{22}/U_{23}									
U_{22}/U_{24}									
U_{22}/U_{25}									
U_{22}/U_{26}									
U_{23}/U_{24}									
U_{23}/U_{25}									
U_{23}/U_{26}									
U_{24}/U_{25}									
U_{24}/U_{26}									
U_{25}/U_{26}									

quality and safety construction, U_{24} ; road construction cost and schedule control, U_{25} ; project operation cost and safety guarantee, U_{26} .

4. At the level of legal responsibility for large municipal road projects (U_3), please compare the relative importance of the following secondary indicators two by two. Information disclosure, U_{31} ; effective regulation, U_{32} ; design, construction, and operation in accordance with transportation industry norms and legal requirements, U_{33} .

	Extremely Important	Very Important	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{31}/U_{32}									
U_{31}/U_{33}									
U_{32}/U_{33}									

5. At the level of environmental and ethical responsibility of large municipal road projects (U_4), please compare the relative importance of the following secondary indicators two by two. Water, noise, dust, and other pollution control, U_{41} ; environmental, ecological, and cultural protection along the road, U_{42} ; consideration of the impact of the proposed road on the existing tracks, U_{43} ; interoperability with the surrounding environment, U_{44} ; protection of the rights and interests of the participating employees, U_{45} ; coordination of the road construction and pipeline construction sequence, U_{46} ; awareness of environmental protection, U_{47} ; monitoring and prosecution of illegal acts, U_{48} .

	Extremely Important	Very Important	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{41}/U_{42}									
U_{41}/U_{43}									
U_{41}/U_{44}									
U_{41}/U_{45}									

	Extremely Important	Very Important	Obviously Important	Slightly More Important	Equally Important	Slightly Unimpor- tant	Obviously Not Important	Very Unimpor- tant	Extremely Unimpor- tant
U_{41}/U_{46}									
U_{41}/U_{47}									
U_{41}/U_{48}									
U_{42}/U_{43}									
U_{42}/U_{44}									
U_{42}/U_{45}									
U_{42}/U_{46}									
U_{42}/U_{47}									
U_{42}/U_{48}									
U_{43}/U_{44}									
U_{43}/U_{45}									
U_{43}/U_{46}									
U_{43}/U_{47}									
U_{43}/U_{48}									
U_{44}/U_{45}									
U_{44}/U_{46}									
U_{44}/U_{47}									
U_{44}/U_{48}									
U_{45}/U_{46}									
U_{45}/U_{47}									
U_{45}/U_{48}									
U_{46}/U_{47}									
U_{46}/U_{48}									
U_{47}/U_{48}									

References

- 1. Flyvbjerg, B. What You Should Know About Megaprojects and Why: An Overview. Proj. Manag. J. 2014, 45, 6–19. [CrossRef]
- Lin, H.; Zeng, S.; Ma, H.; Zeng, R.; Tam, V. An indicator system for evaluating megaproject social responsibility. *Int. J. Proj. Manag.* 2017, 35, 1415–1426. [CrossRef]
- Nilashi, M.; Rupani, P.F.; Rupani, M.M.; Kamyab, H.; Shao, W.; Ahmadi, H.; Rashid, T.A.; Aljojo, N. Measuring sustainability through ecological sustainability and human sustainability: A machine learning approach. *J. Clean. Prod.* 2019, 240, 118162. [CrossRef]
- 4. Miller, R.; Hobbs, B. Governance Regimes for Large Complex Projects. Proj. Manag. J. 2005, 36, 42–50. [CrossRef]
- Flyvbjerg, B. Over Budget, Over Time, Over and Over Again: Managing Major Projects. In *The Oxford Handbook of Project Management*; Oxford University Press: Oxford, UK, 2011; pp. 321–344. [CrossRef]
- 6. Demetriades, P.O.; Mamuneas, T.P. Intertemporal output and employment effects of public infrastructure capital: Evidence from 12 OECD economies. *Econ. J.* 2000, *110*, 687–712. [CrossRef]
- Wang, G.; Wu, P.; Wu, X.; Zhang, H.; Guo, Q.; Cai, Y. Mapping global research on sustainability of megaproject management: A scientometric review. J. Clean. Prod. 2020, 259, 120831. [CrossRef]
- 8. Van Marrewijk, A.; Clegg, S.R.; Pitsis, T.S.; Veenswijk, M. Managing public-private megaprojects: Paradoxes, complexity, and project design. *Int. J. Proj. Manag.* 2008, 26, 591–600. [CrossRef]
- 9. Davies, A.; Gann, D.; Douglas, T. Innovation in Megaprojects: Systems Integration at London Heathrow Terminal 5. *Calif. Manag. Rev.* **2009**, *51*, 101–125. [CrossRef]
- 10. Lin, H.; Zeng, S.; Ge, S.; Chen, Y. Can the bullet train speed up climate change mitigation in China? *Front. Eng. Manag.* 2017, *4*, 104–105. [CrossRef]
- 11. Faber, B. Trade Integration, Market Size, and Industrialization: Evidence from China's National Trunk Highway System. *Rev. Econ. Stud.* **2014**, *81*, 1046–1070. [CrossRef]
- 12. Shen, L.; Wu, Y.; Zhang, X. Key Assessment Indicators for the Sustainability of Infrastructure Projects. *J. Constr. Eng. Manag.* 2011, 137, 441–451. [CrossRef]

- 13. He, Q.; Chen, X.; Wang, G.; Zhu, J.; Li, Y. Managing social responsibility for sustainability in megaprojects: An innovation transitions perspective on success. *J. Clean. Prod.* **2019**, 241, 118395. [CrossRef]
- Zeng, S.X.; Ma, H.Y.; Lin, H.; Zeng, R.C.; Tam, V. Social responsibility of major infrastructure projects in China. Int. J. Proj. Manag. 2015, 33, 537–548. [CrossRef]
- 15. Li, X.; Gao-Zeller, X.; Rizzuto, T.E.; Yang, F. Institutional pressures on corporate social responsibility strategy in construction corporations: The role of internal motivations. *Corp. Soc. Responsib. Environ. Manag.* **2019**, *26*, 721–740. [CrossRef]
- 16. Wu, H.; Xue, X.; Zhao, Z.; Wang, Z.; Shen, G.Q.; Luo, X. Major Knowledge Diffusion Paths of Megaproject Management: A Citation-Based Analysis. *Proj. Manag. J.* 2020, *51*, 242–261. [CrossRef]
- 17. Derakhshan, R.; Turner, R.; Mancini, M. Project governance and stakeholders: A literature review. *Int. J. Proj. Manag.* 2019, 37, 98–116. [CrossRef]
- Liu, Z.; Wang, L.; Sheng, Z.; Gao, X. Social responsibility in infrastructure mega-projects: A case study of ecological compensation for Sousa chinensis during the construction of the Hong Kong-Zhuhai-Macao Bridge. *Front. Eng. Manag.* 2018, *5*, 98–108. [CrossRef]
- Tang, F.Y. Study on Construction Technology of Municipal Road and Bridge Concrete. In Proceedings of the 3rd International Conference on Energy Equipment Science and Engineering (ICEESE), Beijing, China, 28–31 December 2017.
- 20. Rokicki, B.; Stepniak, M. Major transport infrastructure investment and regional economic development—An accessibility-based approach. *J. Transp. Geogr.* 2018, 72, 36–49. [CrossRef]
- Amekudzi, A.A.; Thomas-Mobley, L.; Ross, C. Transportation planning and infrastructure delivery in major cities and megacities. *Transp. Res. Rec.* 2007, 1997, 17–23. [CrossRef]
- 22. Merrow, E.W.; McDonnell, L.M.; Arguden, R.Y. Understanding the Outcomes of Mega-Projects: A Quantitative Analysis of Very Large Civilian Projects; RAND Corporation: Santa Monica, CA, USA, 1988.
- 23. Huang, J. Feasibility Analysis on the Application of 3D Printing Technology on Municipal Roads. In *Journal of Physics: Conference Series;* IOP Publishing: Chengdu, China, 2020; p. 072023. [CrossRef]
- 24. Shaofu, L. The application of EPC general contracting model in municipal road engineering. *IOP Conf. Ser. Mater. Sci. Eng.* 2019, 688, 055066. [CrossRef]
- Ugwu, O.O.; Kumaraswamy, M.M.; Wong, A.; Ng, S.T. Sustainability appraisal in infrastructure projects (SUSAIP): Part 2: A case study in bridge design. *Autom. Constr.* 2006, 15, 239–251. [CrossRef] [PubMed]
- Gil, N.; Beckman, S. INTRODUCTION: Infrastructure Meets Business: Building new bridges, mending old ones. *Calif. Manag. Rev.* 2009, 51, 6–29. [CrossRef]
- Ma, H.; Zeng, S.; Lin, H.; Chen, H.; Shi, J.J. The societal governance of megaproject social responsibility. *Int. J. Proj. Manag.* 2017, 35, 1365–1377. [CrossRef]
- 28. Stone, R. Three Gorges Dam: Into the unknown. Science 2008, 321, 628–632. [CrossRef] [PubMed]
- Xie, L.; Han, T.; Chu, H.; Xia, B. Behavior Selection of Stakeholders toward Megaproject Social Responsibility: Perspective from Social Action Theory. *Adv. Civ. Eng.* 2019, 2019, 4956067. [CrossRef]
- Ansar, A.; Flyvbjerg, B.; Budzier, A.; Lunn, D. Should we build more large dams? The actual costs of hydropower megaproject development. *Energy Policy* 2014, 69, 43–56. [CrossRef]
- 31. Carroll, A.B. Corporate Social Responsibility Evolution of a Definitional Construct. Bus. Soc. 1999, 38, 268–295. [CrossRef]
- 32. Zhao, Z.Y.; Zhao, X.J.; Davidson, K.; Jian, Z. A corporate social responsibility indicator system for construction enterprises. *J. Clean. Prod.* **2012**, 29–30, 277–289. [CrossRef]
- Lwin, M.; Panuwatwanich, K. Identification and evaluation of green building assessment indicators for myanmar. J. Green Build. 2021, 16, 143–172. [CrossRef]
- Zhou, Z.; Mi, C. Social responsibility research within the context of megaproject management: Trends, gaps and opportunities. *Int. J. Proj. Manag.* 2017, 35, 1378–1390. [CrossRef]
- Adetayo, O.; Innocent, M. Barriers to BIM-Based Life Cycle Sustainability Assessment for Buildings: An Interpretive Structural Modelling Approach. *Buildings* 2022, 12, 324. [CrossRef]
- Wuni, I.Y.; Shen, G.Q.; Osei-Kyei, R. Sustainability of off-site construction: A bibliometric review and visualized analysis of trending topics and themes. J. Green Build. 2020, 15, 131–154. [CrossRef]
- 37. Sang, M.; Zhang, Y.; Ye, K.; Jiang, W. Moderating Effects of Internationalization between Corporate Social Responsibility and Financial Performance: The Case of Construction Firms. *Buildings* **2022**, *12*, 185. [CrossRef]
- 38. Heather, F.R. Corporate social responsibility disclosure: The three concentric circles model—A proposed framework for classifying sustainability initiatives in the fashion supply-chain. *Int. J. Bus. Glob.* **2014**, *13*, 76. [CrossRef]
- 39. Kolk, A. The social responsibility of international business: From ethics and the environment to CSR and sustainable development. *J. World Bus.* **2016**, *51*, 23–34. [CrossRef]
- 40. Tam, V.W.Y.; Tam, C.M.; Zeng, S.X.; Chan, K.K. Environmental performance measurement indicators in construction. *Build*. *Environ.* **2006**, *41*, 164–173. [CrossRef]
- 41. Wood, D.J. Measuring Corporate Social Performance: A Review. Int. J. Manag. Rev. 2010, 12, 50–84. [CrossRef]
- 42. Zheng, S.; Kahn, M.E. China's bullet trains facilitate market integration and mitigate the cost of megacity growth. *Proc. Natl. Acad. Sci. USA* **2013**, *110*, E1248–E1253. [CrossRef] [PubMed]

- 43. Hueting, R.; Reijnders, L. Broad sustainability contra sustainability: The proper construction of sustainability indicators. *Ecol. Econ.* **2004**, *50*, 249–260. [CrossRef]
- Li, H.; Wang, F.; Zhang, C.; Wang, L.; An, X.; Dong, G. Sustainable supplier selection for water environment treatment publicprivate partnership projects. J. Clean. Prod. 2021, 324, 129218. [CrossRef]
- 45. Qiu, W.; Liu, Y.; Lu, F.; Huang, G. Establishing a sustainable evaluation indicator system for railway tunnel in China. *J. Clean. Prod.* **2020**, *268*, 122150. [CrossRef]
- 46. Seager, T.P.; Theis, T.L. A taxonomy of metrics for testing the industrial ecology hypotheses and application to design of freezer insulation. *J. Clean. Prod.* 2004, *12*, 865–875. [CrossRef]
- 47. Erkman, S. Industrial ecology: An historical view. J. Clean. Prod. 1997, 5, 1–10. [CrossRef]
- Bai, C.; Kusi-Sarpong, S.; Ahmadi, H.B.; Sarkis, J. Social sustainable supplier evaluation and selection: A group decision-support approach. Int. J. Prod. Res. 2019, 57, 7046–7067. [CrossRef]
- 49. Azhar, S.; Carlton, W.A.; Olsen, D.; Ahmad, I. Building information modeling for sustainable design and LEED (R) rating analysis. *Autom. Constr.* 2011, 20, 217–224. [CrossRef]
- De Siqueira, A.C.H.; Mohammad, K.N.; Ahmed, W.A.H.; Assed, H.; Elaine, V. Sustainable Urban Development in Slum Areas in the City of Rio de Janeiro Based on LEED-ND Indicators. *Buildings* 2020, 10, 116. [CrossRef]
- Pushkar, S. LEED-CI V3 and V4 Gold Projects for Office Spaces: The Difference between Shanghai and California. *J. Green Build.* 2021, 16, 29–44. [CrossRef]
- 52. Pushkar, S. The leed-commercial interiors (V4) projects in california. J. Green Build. 2021, 16, 57–72. [CrossRef]
- 53. Hoogmartens, R.; Van Passel, S.; Van Acker, K.; Dubois, M. Bridging the gap between LCA, LCC and CBA as sustainability assessment tools. *Environ. Impact Assess. Rev.* 2014, 48, 27–33. [CrossRef]
- 54. Huang, M.; Dong, Q.; Ni, F.; Wang, L. LCA and LCCA based multi-objective optimization of pavement maintenance. *J. Clean. Prod.* **2021**, *283*, 124583. [CrossRef]
- Cinelli, M.; Coles, S.R.; Kirwan, K. Analysis of the potentials of multi criteria decision analysis methods to conduct sustainability assessment. Ecol. Indic. 2014, 46, 138–148. [CrossRef]
- Joseph, S. Making a Sustainability Business Case for Alternative Building Designs Using the LEED Requirements. J. Green Build. 2006, 1, 58–66. [CrossRef]
- 57. Marwa, M.G.M.; Zeyad, M.T.E.S.; Ahmed AbdelMonteleb, M.A.; Mona, G.I. Assessment of Green Building Materials' Attributes to Achieve Sustainable Building Façades Using AHP. *Buildings* **2021**, *11*, 474. [CrossRef]
- Lyu, H.-M.; Sun, W.-J.; Shen, S.-L.; Zhou, A.-N. Risk Assessment Using a New Consulting Process in Fuzzy AHP. J. Constr. Eng. Manag. 2020, 146, 04019112. [CrossRef]
- 59. Kubler, S.; Robert, J.; Derigent, W.; Voisin, A.; Le Traon, Y. A state-of the-art survey & testbed of fuzzy AHP (FAHP) applications. *Expert Syst. Appl.* **2016**, *65*, 398–422. [CrossRef]
- 60. Usama, I.; Fam, S.; Yehia, M.; Muwaffaq, A.; Emad, O. Hybrid AHP-Fuzzy TOPSIS Approach for Selecting Deep Excavation Support System. *Buildings* **2022**, *12*, 295. [CrossRef]
- 61. Marcus, V.A.P.M.F.; Bruno, B.F.d.C.; Mohammad, N.; Karoline, V.F.; de Marcos Barreto, M.; Assed, N.H. Sustainability Assessment of a Low-income Building: A BIM-LCSA-FAHP-based Analysis. *Buildings* **2022**, *12*, 181. [CrossRef]
- 62. John, E. Accounting for the triple bottom line. Meas. Bus. Excell. 1998, 2, 18–22. [CrossRef]
- 63. Liu, X.; Liu, H.; Chen, J.; Liu, T.; Deng, Z. Evaluating the sustainability of marine industrial parks based on the DPSIR framework. *J. Clean. Prod.* **2018**, *188*, 158–170. [CrossRef]
- 64. Kahraman, C.; Cebeci, U.; Ulukan, Z. Multi-criteria supplier selection using fuzzy AHP. *Logist. Inf. Manag.* 2003, 16, 382–394. [CrossRef]
- 65. Lee, A.H.I.; Chen, W.C.; Chang, C.J. A fuzzy AHP and BSC approach for evaluating performance of IT department in the manufacturing industry in Taiwan. *Expert Syst. Appl.* **2008**, *34*, 96–107. [CrossRef]
- Hu, Y.; Chan, A.P.C.; Le, Y.; Xu, Y.; Shan, M. Developing a Program Organization Performance Index for Delivering Construction Megaprojects in China: Fuzzy Synthetic Evaluation Analysis. J. Manag. Eng. 2016, 32, 05016007. [CrossRef]
- 67. Li, T.H.Y.; Ng, S.T.; Skitmore, M. Evaluating stakeholder satisfaction during public participation in major infrastructure and construction projects: A fuzzy approach. *Autom. Constr.* **2013**, *29*, 123–135. [CrossRef]
- Qh, A.; Lan, L.A.; Yi, H.B.; Apcc, C. Measuring the complexity of mega construction projects in China—A fuzzy analytic network process analysis. *Int. J. Proj. Manag.* 2015, 33, 549–563.
- 69. Ting-Ya, H.; Shih-Tong, L.; Gwo-Hshiung, T. Fuzzy MCDM approach for planning and design tenders selection in public office buildings. *Int. J. Proj. Manag.* 2004, 22, 573–584. [CrossRef]
- Becker, A.K.; Ross, B.E.; Albright, D. Evaluating the weighted-sum approach for measuring buildings' adaptability. J. Green Build. 2020, 15, 37–54. [CrossRef]
- Xu, S.; Yang, R. Indigenous Characteristics of Chinese Corporate Social Responsibility Conceptual Paradigm. J. Bus. Ethics 2010, 93, 321–333. [CrossRef]
- Zhai, Z.; Ahola, T.; Le, Y.; Xie, J. Governmental Governance of Megaprojects: The Case of EXPO 2010 Shanghai. *Proj. Manag. J.* 2017, 48, 37–50. [CrossRef]
- Locatelli, G.; Mariani, G.; Sainati, T.; Greco, M. Corruption in public projects and megaprojects: There is an elephant in the room! Int. J. Proj. Manag. 2017, 35, 252–268. [CrossRef]

- 74. Yang, D.; He, Q.; Cui, Q.; Hsu, S.-C. Non-economic motivations for organizational citizenship behavior in construction megaprojects. *Int. J. Proj. Manag.* 2020, *38*, 64–74. [CrossRef]
- 75. Wang, G.; Locatelli, G.; Wan, J.; Li, Y.; Le, Y. Governing behavioral integration of top management team in megaprojects: A social capital perspective. *Int. J. Proj. Manag.* 2021, *39*, 365–376. [CrossRef]
- 76. Lin, H.; Zeng, S.; Ma, H. Water scheme acts as ecological buffer. Nature 2016, 529, 283. [CrossRef] [PubMed]
- 77. Levitt, R.E. CEM research for the next 50 years: Maximizing economic, environmental, and societal value of the built environment. *J. Constr. Eng. Manag.* 2007, 133, 619–628. [CrossRef]
- Oladinrin, T.O.; Ho, C.M.-F. Strategies for Improving Codes of Ethics Implementation in Construction Organizations. *Proj. Manag. J.* 2014, 45, 15–26. [CrossRef]
- 79. Darendeli, I.S.; Hill, T.L. Uncovering the complex relationships between political risk and MNE firm legitimacy: Insights from Libya. *J. Int. Bus. Stud.* **2016**, *47*, 68–92. [CrossRef]
- Ma, H.; Sun, D.; Zeng, S.; Lin, H.; Shi, J.J. The Effects of Megaproject Social Responsibility on Participating Organizations. *Proj. Manag. J.* 2021, 52, 418–433. [CrossRef]
- Ke, Y.; Wang, S.; Chan, A.P.C.; Lam, P.T.I. Preferred risk allocation in China's public-private partnership (PPP) projects. *Int. J. Proj. Manag.* 2010, 28, 482–492. [CrossRef]
- 82. Zhai, Z.; Shan, M.; Le, Y. Investigating the impact of governmental governance on megaproject performance: Evidence from China. *Technol. Econ. Dev. Econ.* 2020, 26, 449–478. [CrossRef]
- Vanessa, V.; Susan, M.B. Assessing the link between public opinion and social sustainability in building and infrastructure projects. J. Green Build. 2015, 10, 177–190. [CrossRef]
- Mengmeng, W.; Lili, L.; Caixia, H.; Xiaotong, G.; Hanliang, F. Building and Health: Mapping the Knowledge Development of Sick Building Syndrome. *Buildings* 2022, 12, 287. [CrossRef]
- 85. Zhou, Z.; Zhou, X.; Qian, L. Online Public Opinion Analysis on Infrastructure Megaprojects: Toward an Analytical Framework. J. Manag. Eng. 2021, 37, 04020105. [CrossRef]
- 86. Lin, H.; Sui, Y.; Ma, H.; Wang, L.; Zeng, S. CEO Narcissism, Public Concern, and Megaproject Social Responsibility: Moderated Mediating Examination. *J. Manag. Eng.* **2018**, *34*, 04018018. [CrossRef]
- 87. Flyvbjerg, B. Why Mass Media Matter to Planning Research: The Case of Megaprojects. J. Plan. Educ. Res. 2012, 32, 169–181. [CrossRef]
- 88. Toor, S.-U.-R.; Ogunlana, S.O. Beyond the 'iron triangle': Stakeholder perception of key performance indicators (KPIs) for large-scale public sector development projects. *Int. J. Proj. Manag.* 2010, *28*, 228–236. [CrossRef]
- 89. Flyvbjerg, B.; Holm, M.S.; Buhl, S. Underestimating costs in public works projects-Error or lie? *J. Am. Plan. Assoc.* 2002, 68, 279–295. [CrossRef]
- El Mekaoui, A.; Tariq, R.; Ramirez, O.B.; Mendez-Monroy, P.E. Sustainability, Sociocultural Challenges, and New Power of Capitalism for Renewable Energy Megaprojects in an Indigenous Mayan Community of Mexico. *Sustainability* 2020, 12, 7432. [CrossRef]
- 91. Cantarelli, C.C.; Molin, E.J.E.; van Wee, B.; Flyvbjerg, B. Characteristics of cost overruns for Dutch transport infrastructure projects and the importance of the decision to build and project phases. *Transp. Policy* **2012**, *22*, 49–56. [CrossRef]
- 92. Cantarelli, C.C.; Flyvbjerg, B.; Molin, E.J.E.; van Wee, B. Cost Overruns in Large-scale Transportation Infrastructure Projects: Explanations and Their Theoretical Embeddedness. *Eur. J. Transp. Infrastruct. Res.* **2010**, *10*, 5–18.
- Bowen, P.; Akintoye, A.; Pearl, R.; Edwards, P.J. Ethical behaviour in the South African construction industry. *Constr. Manag. Econ.* 2007, 25, 631–648. [CrossRef]
- 94. Lin, H.; Zeng, S.; Ma, H.; Chen, H. Does commitment to environmental self-regulation matter? An empirical examination from China. *Manag. Decis.* 2015, 53, 932–956. [CrossRef]
- 95. Shen, L.-Y.; Tam, V.W.Y.; Tam, L.; Ji, Y.-B. Project feasibility study: The key to successful implementation of sustainable and socially responsible construction management practice. *J. Clean. Prod.* **2010**, *18*, 254–259. [CrossRef]
- Babatunde, S.O.; Opawole, A.; Akinsiku, O.E. Critical success factors in public-private partnership (PPP) on infrastructure delivery in Nigeria. J. Facil. Manag. 2012, 10, 212–225. [CrossRef]
- 97. Li, Y.; Lu, Y.; Cui, Q.; Han, Y. Organizational Behavior in Megaprojects: Integrative Review and Directions for Future Research. *J. Manag. Eng.* **2019**, *35*, 04019009. [CrossRef]